Author’s response to reviews

Title: Health-related quality of life in type-2 diabetes patients: a cross-sectional study in East China

Authors:
You Lu (ly012202@163.com)
Ningjian Wang (wnj486@126.com)
Yi Chen (chenyi9h@126.com)
Xiaomin Nie (niexiaomin2016@163.com)
Qin Li (liqin20001@163.com)
Bing Han (foxhb1423@126.com)
Yingchao Chen (ccyanzhi@126.com)
Fangzhen Xia (xiafangzhen@126.com)
Zhen Cang (cangzhenfly@yeah.net)
Meng Lu (rubylumeng@yahoo.com.cn)
Ying Meng (mengying_1988@163.com)
Yingli Lu (luyingli2008@126.com)

Version: 3 Date: 19 May 2017

Author’s response to reviews:

Point-by-point response letter

Dear Editor and Reviewers:

We are truly grateful to yours and other reviewers’ critical comments and thoughtful suggestions. We feel lucky that our manuscript went to these reviewers as the valuable comments from them. Based on these comments and suggestions, we have made careful modification on the original manuscript. All changes made to the text are indicated in the text by using track changes. We look forward to hearing from you soon for a favorable decision.
Below you will find our point-by-point responses to the reviewers’ comments/questions:

1. First of all, we had no change in the order of authors. The order in the title page was correct. The order in the system was an error. We were sorry for that and made it right this time.

2. As suggested, we have represented author’s names by their full initials. We made an error about Chi Chen and Yi Chen in the Authors’ contributions section. It was Yi Chen, not Chi Chen, who contributed to this article. We also provided more justification for the contributions.

3. As suggested by editors, we have removed the figure legend embedded within the Figure 1 file and placed it at the end of the main manuscript.

4. We have updated all the numbers in the title page.

5. This time we have submitted a clean copy as you requested.

Thank you again for your time and consideration. We hope that our manuscript could be considered for publication in your journal.

Best wishes,

You Lu